Author's response to reviews

Title: Study Protocol: A pragmatic randomised controlled trial of a 12-week physical activity and nutritional education program for overweight Aboriginal and Torres Strait Islander Women

Authors:

Karla J Canuto (Karla.Canuto@postgrads.unisa.edu.au)
Robyn A McDermott (Robyn.McDermott@unisa.edu.au)
Margaret Cargo (Margaret.Cargo@unisa.edu.au)
Adrian J Esterman (Adrian.Esterman@unisa.edu.au)

Version: 3 Date: 3 August 2011

Author's response to reviews:

3 August 2011

Dear Editor of BMC Public Health

Please consider this paper for publication in the Biomedical Journal of Public Health.

We believe that the Biomedical Journal of Public Health is the most appropriate place to publish this works as it is open access so that other researchers and practitioners can freely access this paper.

Yours sincerely

Karla Canuto
PhD Candidate
University of South Australia
Adelaide